# THE ILLUSION OF STAKEHOLDER GOVERNANCE: CORPORATE PURPOSE OR CORPORATE SPIN?

Fernando Camastra \*, Giacinto Coniglio \*\*, Antonio Blasotti \*\*\*

\*Corresponding author, Department of Management, Sapienza University of Rome, Rome, Italy
Contact details: Department of Management, Sapienza University of Rome, Via del Castro Laurenziano 9, 00153 Rome, Italy

\*\*Department of Management, Sapienza University of Rome, Rome, Italy

\*\*\* Department of Law Studies, University of Salento, Lecce, Italy



How to cite this paper: Camastra, F., Coniglio, G., & Blasotti, A. (2025). The illusion of stakeholder governance: Corporate purpose or corporate spin? Corporate Governance and Organizational Behavior Review, 9(2), 20–33. https://doi.org/10.22495/cgobrv9i2p2

Copyright © 2025 The Authors

This work is licensed under a Creative Commons Attribution 4.0 International License (CC BY 4.0). https://creativecommons.org/licenses/by/

ISSN Online: 2521-1889 ISSN Print: 2521-1870

**Received:** 23.06.2025 **Revised:** 06.08.2025; 26.09.2025

Accepted: 03.10.2025

JEL Classification: D21, G34, K22, L21, M14

**DOI:** 10.22495/cgobrv9i2p2

# **Abstract**

emphasis on stakeholder governance The growing fundamentally reshaped the corporate governance debate, challenging the traditional focus on shareholder primacy. The 2019 Business Roundtable (BRT) Statement marked a pivotal turn by committing signatory firms to consider the interests of all stakeholders. Despite this public pledge, substantial doubts remain about the depth and sincerity of corporate adoption (Bebchuk & Tallarita, 2020). This study assesses whether the 39 publicly listed companies that endorsed the BRT Statement between 2019 and 2024 have implemented concrete governance reforms. Through an empirical analysis of each firm's published governance guidelines, examining board composition, committee charters, and reporting disclosures, we track changes over a five-year period. Our results reveal a significant divergence between proclaimed intentions and actual practice: most firms continue to anchor their governance structures in shareholder value, while only a minority exhibit genuine steps toward embedding stakeholder interests. Future research should integrate quantitative data on corporate behavior and stakeholder outcomes to provide a more comprehensive evaluation. Overall, this paper offers an updated empirical appraisal of whether leading BRT signatories have translated stakeholderism from aspirational rhetoric governance reality.

**Keywords:** Shareholder Primacy, Stakeholder, Governance, Regulation, Business Roundtable

**Authors' individual contribution:** Conceptualization — F.C. and G.C.; Methodology — F.C. and G.C.; Validation — F.C., G.C., and A.B.; Formal Analysis — F.C. and G.C.; Investigation — F.C. and G.C.; Data Curation — F.C. and G.C.; Writing — F.C., G.C., and A.B.; Visualization — F.C. and G.C.

**Declaration of conflicting interests:** The Authors declare that there is no conflict of interest.

## 1. INTRODUCTION

In recent years, stakeholder governance has assumed an increasingly central role in academic debate and public policies related to corporate regulation (Freeman et al., 2020). The growing attention to the impact of companies on communities, workers, and other stakeholders has led to an in-depth discussion of the limitations and potential of stakeholderism (Amis et al., 2020).

The concept of stakeholder-oriented governance has been presented as an alternative to the traditional doctrine of shareholder primacy, suggesting that companies should assume broader responsibilities toward all stakeholders, not just their investors (Schaltegger et al., 2019; Shah & Li, 2025). However, while many scholars and policymakers view stakeholderism as a fairer and more sustainable model (Quinto et al., 2019), others raise doubts about its actual implementation and the sincerity of companies in adhering to their

stated commitments (Yoshikawa et al., 2021). In this scenario, the 2019 Business Roundtable (BRT) Statement marked a key moment, with over 180 chief executive officers (CEOs) of major companies declaring their commitment to more inclusive governance (Esposito De Falco, The CEOs pledged to adopt a set of guidelines for sustainable corporate behavior, rejecting the idea of a company based solely on profit for shareholders. The document sought to mark the end of shareholder primacy in favor of stakeholder primacy. However, fundamental questions remain regarding the actual operational and structural transformation of the participating companies. On one hand, the 2019 BRT Statement was welcomed as a significant step toward this transformation (Bebchuk & Tallarita, 2020); on the other hand, some studies question the real effectiveness of such declarations. Specifically, Bebchuk and Tallarita (2020) highlight how BRT signatory companies have not effectively changed their governance guidelines, maintaining a clear centrality of shareholders in strategic decisions. Additionally, an analysis of corporate documents (such as corporate governance guidelines, requests for U.S. Securities and Exchange Commission (SEC) no-action letters, bylaws, proxy statements, and director compensation principles) has shown that companies have systematically rejected shareholder proposals aimed at strengthening stakeholder-oriented governance and have not adopted stakeholder-related performance metrics.

This discrepancy between statements and actions raises critical questions about the nature genuinely stakeholderism: are companies implementing a more inclusive governance model, or is it merely a rhetorical mechanism aimed at enhancing corporate reputation and reducing regulatory pressures? The academic literature on corporate governance has increasingly focused on stakeholderism, analyzing its potential benefits and risks (Bebchuk & Tallarita, 2020; Freeman et al., 2021). The academic debate is divided between stakeholderism advocates, who see its potential to mitigate corporate negative externalities, and critics who consider it a primarily rhetorical phenomenon, lacking substantive operational change. Bebchuk and Tallarita's (2020) empirical analysis revealed that in the two years following the BRT Statement, signatory companies did not implement substantial governance changes, instead reaffirming shareholder centrality in their official documents. However, the regulatory and social context is constantly evolving, and there is a need to update this evidence in light of more recent developments.

The lack of substantial changes in governance structures suggests that many companies may be using stakeholderism language to respond to growing social expectations without, however, translating these commitments into concrete actions (Dmytryev et al., 2021). Examining this issue is crucial, as corporate governance not only influences wealth distribution and social well-being but also determines capital allocation strategies and investment approaches (Attanasio et al., 2022). If stakeholderism were genuinely adopted as a guiding principle, we should observe a redefinition of corporate objectives, integrating metrics that consider value creation for all stakeholders, not just shareholders (Zaman et al., 2022).

In light of the latest evidence and emerging challenges, our study aims to verify whether, drawing inspiration from the empirical analysis conducted by Bebchuk and Tallarita (2020) with

updated data from 2024, the companies that signed the BRT Statement have effectively modified their governance practices toward a stakeholder-oriented model. Our goal is to analyze whether large corporations are genuinely shifting toward stakeholderism or if, instead, traditional governance models persist. This study will help clarify whether stakeholderism is evolving into a consolidated practice or remains a mere communication tool with no concrete impact on corporate governance. Thus, our research seeks to answer the following research question:

aut signed modified thoir RQ: Have the companies BRT Statement effectively governance. practices corporate reflect to stakeholder-oriented model, or does their adherence to the BRT remain purely symbolic, and what is the nature of this evolution?

To address this question, we conduct an empirical analysis based on the collection and review of corporate governance guidelines from the companies in our sample.

Our results indicate that, despite the ongoing debate on stakeholderism, the analyzed companies have not implemented significant structural changes in their governance approach. Corporate guidelines continue to emphasize shareholder primacy, and responses to investor proposals confirm a tendency to maintain a traditional model. Only one of the analyzed companies has explicitly aligned managerial incentives with stakeholder-oriented metrics, demonstrating that this phenomenon remains an exception rather than a widespread practice. These findings suggest that the BRT Statement has had a limited impact on actual corporate practices. This study contributes to the literature on stakeholderism by providing updated empirical analysis of the real implementation of stakeholder-oriented governance in large corporations. Our results highlight that, despite principled declarations, the vast majority of companies have not implemented substantial changes in governance approach. This discrepancy between rhetoric and concrete actions raises crucial questions about the authenticity of stated commitments. Specifically, our study confirms that the governance guidelines of companies that signed the BRT Statement have not undergone structural transformations favoring greater stakeholder involvement in corporate strategies. This suggests that stakeholderism is often used as a marketing tool rather than a genuine evolution of the governance model. Our study has significant implications for policymakers and investors, emphasizing the need for more rigorous measurement tools to assess corporate commitment stakeholders. Finally, we provide a clear framework on the role of institutional investors and regulatory pressures in shaping the evolution of corporate governance, highlighting the need for regulatory reforms that ensure greater accountability and transparency in corporate strategies.

The remainder of this paper is structured as follows. Section 2 reviews the relevant literature shareholder primacy and stakeholder governance, outlining the theoretical foundations and empirical evidence, and formulates the main hypotheses. Section 3 details the research methodology, including the sampling criteria, data collection, and the systematic textual analysis employed. Section 4 presents the empirical results, illustrating the persistence of shareholderism, instances of genuine transition, and the adoption of

hybrid models. Section 5 discusses the theoretical relevance and practical implications of our findings, while Section 6 concludes the paper by summarizing the main contributions, acknowledging limitations, and suggesting avenues for future research.

#### 2. LITERATURE REVIEW

#### 2.1. Shareholder primacy

The debate on corporate governance has historically contrasted fundamental paradigms: 1) shareholder primacy and 2) stakeholder governance. The shareholder perspective, rooted in agency theory, argues that the primary objective of a company is to maximize shareholder value (Jensen & Meckling, 1976). From this standpoint, corporate is measured performance through indicators, while the role of stakeholders is marginal and instrumental in achieving shareholder value (Tirole, 2010). One of the strengths of this paradigm is the clarity of corporate objectives and the ability to assess performance objectively through financial returns and stock value growth (Hart & Zingales, 2017). Moreover, financial markets play an external monitoring role, mitigating the risk of managerial incentivizing misalignment and operational efficiency (Shleifer & Vishny, 1997). However, this view has been criticized, particularly following financial crises and growing pressures for a more sustainable business model. Excessive emphasis on profit maximization has led to opportunistic as production behaviors such cost-cutting, offshoring, and reduced investments in human capital, with long-term negative consequences (Stout, 2024). The short-term orientation has driven many companies to prioritize immediate financial decisions at the expense of sustainable growth strategies (Lazonick, 2014).

In contrast to shareholder primacy, stakeholder theory, formulated by Freeman (2010), proposes a model in which businesses must serve a broader set of stakeholders, including employees, customers, suppliers, and communities. This theory argues that companies do not operate in a social and economic vacuum but are part of an interdependent system where corporate success also depends on stakeholder well-being (Donaldson & Preston, 1995). Shared value creation is thus seen as a key element for corporate sustainability and organizational resilience (Kramer & Porter, 2011).

Recent research has highlighted that a balanced strategy between shareholders and stakeholders not only reduces reputational and regulatory risks but also enhances firms' attractiveness to long-term investors (Baah et al., 2020; Kyaw et al., 2022). However, despite the ongoing debate and the formal adoption of stakeholder-oriented principles by many companies, evidence suggests that the shareholder model remains predominant in practice.

## 2.2. The growing influence of stakeholderism

In recent years, stakeholder-oriented governance has gained relevance due to the evolution of regulations, investor pressure, and society's growing focus on corporate responsibility (Fransen, 2012; Ortas et al., 2019; Endo, 2020). A significant turning point was represented by the 2019 BRT Statement, in which over 180 CEOs acknowledged the need to serve all stakeholders, marking a break from the traditional doctrine of shareholder primacy. However, doubts

remain regarding the actual implementation of these principles and the consistency between public statements and concrete corporate strategies (Bebchuk & Tallarita, 2020). In addition, global asset managers such as BlackRock, Inc. have begun integrating environmental, social, and governance (ESG) criteria into their investment decisions, encouraging companies to improve transparency and accountability (Eccles & Klimenko, 2019).

Yet, numerous studies have highlighted the limitations of the shareholder-oriented model. For instance, Edmans (2011) demonstrated that companies investing in employee well-being achieve higher long-term financial performance, suggesting that value creation for stakeholders is not in conflict with business success. Similarly, Bebchuk and Tallarita (2020) emphasized how an exclusive focus on shareholders can lead to short-termism, neglecting the long-term sustainability of the company.

Another relevant factor is the emergence of ESG metrics as a central tool for evaluating companies. According to an analysis by Gupta and Aggarwal (2024), companies that adopt structured ESG strategies gain better access to capital and reduce the perceived risk among investors. The growing focus on these factors is leading to a redefinition of the very concept of corporate value, which is no longer limited to financial results alone but also includes social and environmental impacts (Heeb et al., 2023). In this context, some companies integrating ESG criteria into managerial performance evaluations, thereby aligning executive incentives with corporate sustainability goals (Jang et al., 2022). The convergence of regulations across different jurisdictions could accelerate the adoption more inclusive governance models while simultaneously reducing greenwashing and managerial opportunism (Eng et al., 2022).

The rise of stakeholderism is not merely a regulatory and market trend but also reflects an evolution in social expectations regarding businesses. The main challenge remains the ability of companies to translate these principles into concrete actions, avoiding the risk of rhetoric operational unsupported real changes. by On the other hand, with the evolution of regulations and the increasing pressure from investors and civil society, it is likely that the stakeholder-oriented paradigm will become a structural element in the corporate governance of the future (Tibiletti et al., 2021; Thrall, 2021). Thus, stricter regulations, investor attention, and the growing importance of ESG metrics are driving many companies toward more inclusive governance models (Eccles & Klimenko, 2019; Heeb et al., 2023).

#### 2.3. Challenges and constraints of stakeholderism

Despite its growing diffusion, stakeholderism is not exempt from criticism. One of the main objections is the lack of clear and measurable objectives, which can lead to ineffective management and reduced managerial accountability (Bebchuk & Tallarita, 2020). Unlike the shareholder-oriented model, which relies on well-defined financial metrics, stakeholder-oriented governance involves greater discretion and ambiguity in defining corporate priorities (Bhandari et al., 2022). Some studies have shown that this uncertainty can make it difficult to implement effective and consistent strategies (Vural-Yavas, 2021).

From an agency theory perspective, the inclusion of multiple stakeholders can dilute

fiduciary responsibility, managers' increasing the risk of opportunistic and self-referential decisions (Easterbrook & Fischel, 1996). The literature highlights the discrepancy between corporate rhetoric promoting stakeholder interests the actual governance practices implemented by firms. This raises the issue of consistency between declared commitments and actual actions, a crucial topic for the future of stakeholder-oriented management strategies (Yoshikawa et al., 2021; Shahzad et al., 2016). Another limitation concerns the complexity of balancing stakeholder interests, which can generate internal conflicts and hinder firms' adaptability (Jensen, 2002; Hall & Vredenburg, 2005; Donaldson & Preston, 1995). The diversity of stakeholder priorities makes it difficult for companies to make effective decisions without compromising operational efficiency. Additionally, recent research has highlighted that companies adopting a multi-stakeholder governance model often struggle to measure and quantify their impact, to challenges in transparency accountability (Dias et al., 2025).

An additional issue concerns the role of regulatory authorities. While many jurisdictions are implementing regulations that encourage a stakeholder-oriented approach, there is also the risk that adopting mandatory standards may lead to excessive bureaucracy and operational inefficiencies as companies must invest significant resources to comply with regulatory requirements, facing high administrative costs and an increasing documentation burden (Freeman et al., 2020; Braun & Busuioc, 2020). Some scholars argue that regulation should strike a balance between flexibility and mandatory requirements so that companies can implement stakeholder strategies without sacrificing competitiveness and operational agility (Collins, Stakeholder-oriented governance presents challenges in terms of innovation and competitiveness (Ostapenko et al., 2024). While some argue that an inclusive approach fosters the creation of shared value, others highlight that excessive attention to the needs of multiple stakeholders can slow decision-making processes and limit firms' agility in highly competitive markets (Martin & Philips, 2022). Although stakeholderism is considered a more inclusive model compared to shareholder primacy, its practical implementation presents multiple obstacles (Sternberg, 1997). The main challenge remains the development of more effective tools and metrics to ensure that multi-stakeholder governance can be implemented without generating efficiently and measurably inefficiencies or internal conflicts (Joshi et al., 2024). In response to these challenges, many companies are seeking to integrate elements of both models to achieve competitive advantages and enhance corporate resilience (Joshi et al., 2024). In this sense, shareholderism and stakeholderism should not be seen as opposing and irreconcilable models but rather as two complementary perspectives that, if properly balanced, can optimize value creation for both shareholders and stakeholders (Esposito De Falco, 2024; Rautenstrauch & Hummel, 2022).

# 2.4. Research hypotheses

To address our *RQ*, we verify three main hypotheses.

Despite increasing pressure and public statements in favour of a more inclusive approach, we expect established governance structures to

maintain the centrality of shareholders' interests, in line with agency theory and traditional for-profit practices:

H1: Firms continue to follow a shareholder-oriented governance model.

The second hypothesis explores the possibility that statements such as the BRT are not merely symbolic but reflect a concrete evolution towards integrating the interests of employees, customers, suppliers, and communities into corporate strategies, driven by regulatory and market factors. As noted, stakeholder-oriented governance presents challenges in terms of innovation and competitiveness, but a balanced approach can optimise value creation.

H2: Companies are progressively adopting a stakeholder-oriented governance model, recognizing that value creation for all stakeholders represents a sustainable competitive strategy in the long term.

Acknowledging the challenges and constraints of an exclusive adherence to a pure model, the third hypothesis suggests that companies might seek a compromise that balances the need to maximise shareholder value with a focus on a wider group of stakeholders.

H3: Companies are increasingly moving toward, or further solidifying, a hybrid governance model in which economic-financial value creation and stakeholder considerations coexist in a strategic, sustainable, and long-term balance, particularly in the post-BRT context.

#### 3. RESEARCH METHODOLOGY

#### 3.1. Research design and sample selection criteria

This study adopts a qualitative research approach to explore the corporate governance orientation of companies with the highest market capitalization within the BRT. Specifically, the analysis aims to determine whether these companies adopt shareholder-oriented or stakeholder-oriented governance model and whether their corporate governance guidelines reflect a linguistic preference for shareholderist or stakeholderist discourse. The qualitative nature of this research is particularly suited to capturing the nuances and subtleties embedded in governance documents, enabling an in-depth analysis of corporate intentions, strategic priorities, and underlying governance philosophies (Jalilvand et al., 2024). The decision to focus on companies with the highest market capitalization among those belonging to the BRT is based on both theoretical and methodological considerations. From a theoretical perspective, larger companies tend to exert a more significant influence on corporate governance norms and best practices, given their visibility, the level of regulation they are subject to, and their impact on financial markets (Becht et al., 2005). These firms often set precedents that smaller organizations may follow, making them an ideal point of observation for assessing the prevailing governance paradigms within the BRT. From a methodological perspective, limiting the sample to companies with the highest market capitalization ensures a manageable yet representative dataset, avoiding the distortive effects arising from the heterogeneity that could emerge when analyzing organizations with very different governance structures and strategic imperatives. Through this

selection, the study maintains analytical rigor and ensures that the results are generalizable to companies of similar scale and influence (Luttmer, 2007). The final sample includes the 39 largest companies by market capitalization that signed the BRT Statement as of December 31, 2024. A minimum market capitalization of \$100 billion was used as a selection criterion. This threshold was chosen to ensure focus on highly influential firms whose governance practices are widely scrutinized and often serve as industry benchmarks, thereby maximizing the relevance and interpretability of the study's findings. 39 companies collectively represent approximately 59% of the total market capitalization of all 182 BRT signatories as of December 31, 2024. Their combined market cap of about \$10.13 trillion, against an estimated aggregate of \$17.28 trillion for the full membership (based on an average estimate of \$50 billion for the remaining 143 firms), demonstrates that this sample captures a highly representative cross-section of the overall stakeholder governance landscape. The sample includes companies from diverse sectors, with the primary industries represented being financial services (10 of 39, ~ 26%), technology (8 of 39, ~ 21%), industrials and aerospace (6 of 39, ~ 15%), healthcare (5 of 39, ~ 13%) consumer goods (3 of 39, ~8%). This distribution broadly aligns with the overall industry representation within the full BRT signatory population, ensuring a representative cross-section of major, albeit mainly based in the U.S., corporations.

Table 1. Selected sample

Company	Market capitalization (in billions of \$)
Abbott	\$224.58
Amazon	\$2.441
American Express	\$216.02
Apple Inc.	\$3.460
Bank of America	\$358.4
BlackRock, Inc.	\$159.54
Bristol-Myers Squibb	\$122.91
Chevron Corporation	\$282.56
Citigroup, Inc.	\$153.32
Comcast Corporation	\$146.29
Exxon Mobil Corporation	\$484.21
Goldman Sachs	\$198.79
Honeywell	\$146.78
Mastercard	\$502.64
Morgan Stanley	\$219.6
PepsiCo	\$212.2
Pfizer Inc.	\$152.22
Qualcomm Incorporated	\$189.22
Salesforce	\$332.17
Progressive Corporation	\$144.64
Texas Instruments	\$170.38
The Coca-Cola Company	\$275.14
The Home Depot Inc.	\$422.05
UPS	\$116.28
Visa Inc.	\$656.04
AT&T Inc.	\$173.21
Boeing	\$131.05
Cisco Systems, Inc.	\$235.3
IBM Corporation	\$207.24
Johnson & Johnson	\$368.1
JPMorgan Chase & Co.	\$748.46
Oracle	\$442.7
Procter & Gamble	\$397.83
Raytheon Technologies Corporation (RTX Corporation)	\$166.44
S&P Global Inc.	\$161.59
Stryker	\$152.45
Union Pacific	\$154.24
Walmart, Inc.	\$782.45
Lockheed Martin Corporation	\$119.39

Source: Authors' elaboration.

The selection of this sample is not arbitrary but represents a deliberate choice aimed at maximizing the relevance and interpretability of the study's findings. These companies collectively account for a significant portion of the market, and their governance choices often attract the attention of investors, policymakers, and academics.

corporate governance Since guidelines represent formalized expressions of a company's governance philosophy, they serve as an ideal data source for analyzing the governance orientation of these firms. Their standardization across companies enables a systematic comparative analysis. strengthening the robustness of the research design. While this study employs a qualitative research design, its analytical framework is implicitly informed by established comparative corporate governance principles. Notably, the Organization for Economic Co-operation and Development (OECD, 2004) Principles of Corporate Governance and the comprehensive leximetric analysis framework developed by Samanta (2019) provide a rigorous conceptual basis for understanding corporate governance evolution. These frameworks emphasize key measurable factors related to shareholder rights, anti-managerial rights, minority rights, and anti-stakeholder rights. Our qualitative approach, focusing on textual analysis of official guidelines, aligns with the broader conceptualizations of corporate governance evolution found in this literature by identifying explicit and implicit governance commitments, particularly concerning shareholder control over executive remuneration, the market for corporate control, and the presence (or absence) of stakeholder representation and remedies.

#### 3.2. Data collection

To ensure the reliability and authenticity of the dataset, only official corporate governance guidelines were collected directly the companies' official websites. Each document was retrieved from the dedicated "investor relations" or "corporate governance" sections of the respective company's official website. This methodological choice minimizes the risk of issues related to document credibility, ensuring that the analyzed texts reflect the companies' official governance positions rather than third-party interpretations. Moreover, by relying exclusively on primary sources, the study maintains a high degree of methodological transparency and replicability.

collection process The data involved the systematic download of the most recent versions of corporate governance guidelines available at the time of the study. Furthermore, to assess the evolution of governance structures, corporate governance guidelines were collected for multiple discrete time points: 2019 (representing the pre-BRT statement baseline), and then annually or biennially from 2020 to 2024 (post-BRT statement), specifically focusing on documents published in 2020, 2022, and 2024, or the most recent version available for that period if annual updates were not consistently provided. This approach yielded a total of 117 documents for analysis, ensuring a sufficient temporal scope to track changes following the BRT statement and justify the sufficiency of the sample size for a qualitative analysis of this nature. Since governance policies are subject to periodic revisions, particular attention was given to ensuring that only the most up-to-date versions for each time stop were analyzed, capturing the latest governance orientations. To further validate the integrity of the dataset, each document was cross-checked with publicly available governance reports to confirm its consistency.

#### 3.3. Data analysis and validation

The primary analytical approach adopted in this study is a systematic textual analysis of corporate governance guidelines. The textual analysis was conducted on the full text of each corporate governance guideline document, ensuring that all explicit and implicit governance commitments were captured, rather than being limited to specific sections like mission statements. This methodological choice enables an in-depth investigation of both explicit governance commitments and the underlying linguistic frameworks. The analysis was conducted in multiple stages to ensure accuracy and interpretative depth.

First, each corporate governance guideline was examined to determine whether it articulated a governance orientation aligned with a shareholder-centric or stakeholder-centric model. This assessment was carried out by identifying key textual indicators that signal governance priorities. A shareholder-oriented approach was inferred from explicit references to shareholder primacy, fiduciary responsibilities towards shareholder primacy, fiduciary responsibilities towards shareholder value. Conversely, a stakeholder-oriented approach was identified through references to the broader responsibilities of the company towards employees, customers, communities, and other non-shareholder stakeholders.

To systematically code these orientations, predefined linguistic markers were used as analytical heuristics. For example, expressions such as "the primary responsibility of the board... is to oversee the affairs of the company in the interest of shareholders" or "maximize shareholder value in the long term" were classified as indicative of a shareholder governance framework. Conversely, statements emphasizing commitments to "all stakeholders" or the integration of ESG considerations into corporate decision-making were categorized as representative of a stakeholder-oriented model.

In cases where governance orientations were explicitly stated, an additional analysis was conducted to determine implicit governance commitments. This process involved a second review of the guidelines to identify structural or thematic elements that could provide indirect evidence of governance orientation. Furthermore, to enhance the reliability of the classification process, ambiguous cases underwent a review process in which each study author independently assessed the governance orientation of the company under examination. In cases of discrepancies, a deliberative process was adopted, where the authors engaged in structured discussions until a majority decision was reached on classification. Through the identification of key terms, recurring expressions, and specific linguistic structures, it was possible to classify the language adopted by companies into two main categories: 1) shareholder-oriented and 2) stakeholderoriented. A shareholder-oriented language characterized by the use of terms focused on

shareholder value creation, market efficiency, maximizing value for stockholders, and alignment of management and shareowner interests. Expressions such as "maximization of shareholder value", "performance-based compensation", "investor confidence", and "competitiveness in the global market" are typically associated with this perspective.

Conversely, a stakeholder-oriented language places greater emphasis on corporate social responsibility, sustainability, and the creation of value for a broader range of stakeholders, including employees, customers, local communities, and the environment. Terms such as "commitment to the community", "long-term sustainability", "employee well-being", "shared value", and "environmental responsibility" indicate a more inclusive approach toward stakeholders. The labeling protocol for classifying governance orientation (shareholderoriented, stakeholder-oriented, or hybrid) involved a multi-stage process of systematic textual analysis. First, predefined linguistic markers and thematic indicators were used to identify explicit references to governance priorities (e.g., "maximize shareholder value", "fiduciary responsibilities towards shareholders" for shareholder-oriented; "commitment stakeholders". "integration of considerations" for stakeholder-oriented). Second, implicit governance commitments were inferred from structural or thematic elements when explicit statements were absent. For instance, the presence of concrete mechanisms such as stakeholder board representation, stakeholder-linked performance explicit revisions to executive metrics. or compensation structures tied to broader social and environmental outcomes was considered a strong indicator of a genuine stakeholder-orientation. This allowed for the identification of "hybrid" models where both orientations coexisted significantly. The classification was not based on a mere net count of terms but on a qualitative assessment of the dominant philosophy, explicit mandates, and the presence of operationalized mechanisms within the document. In cases where and stakeholder language shareholder significantly present and mutually reinforcing through concrete mechanisms, the document was classified as "hybrid". Where one clearly dominated, without such operationalized mechanisms for the other, it was classified accordingly.

To ensure the reliability and validity of the qualitative research, several strategies were implemented throughout the research process. The use of predefined textual indicators improved the consistency of the classification process, reducing the inherent subjectivity of qualitative textual analysis. Additionally, the involvement of multiple researchers in the verification process strengthened inter-coder reliability, mitigating potential interpretative biases.

Our methodological approach not only allows for a precise classification of governance guidelines but also provides a robust justification for our selective focus. From a scientific perspective, this methodology aligns with the need to ensure that data are not interpreted outside the specific or current context of the company and provides an empirical basis for drawing meaningful conclusions about governance trends that could have a decisive influence on future regulatory and strategic developments in the field of global corporations.

#### 4. RESEARCH RESULTS

# 4.1. Persistence of shareholderism and limited transition towards stakeholderism

The analysis of corporate governance guidelines of the companies in our sample highlights a complex dynamic between shareholder-oriented and stakeholder-oriented approaches. This study has allowed us to identify the evolution of the language and strategic orientation adopted by firms in terms of governance.

From the analysis, it emerges that 97.4% (38 out of 39) of companies continue to operate under a shareholder-oriented governance model, despite their adherence to the BRT and public statements suggesting a greater stakeholder orientation. Companies such as Apple Inc., Morgan Stanley, Abbott Laboratories, and The Home Depot, Inc. maintain a primarily shareholder-focused perspective, as evidenced by the wording of their corporate governance guidelines. For instance, Apple Inc. (2025) explicitly states: "The Board oversees the Chief Executive Officer (the "CEO") and other senior management in the competent and ethical operation of the Corporation on a day-to-day basis and seeks to ensure that the long-term interests of shareholders are being served" (p. 1). Similarly, Morgan Stanley (2025) declares: "Directors should also have the ability to represent

the interests of shareholders" (para. 1). These statements reinforce the persistence of the shareholder model in numerous companies.

A notable example is Honeywell International Inc. (2025), which, despite having updated its governance multiple times, has guidelines consistently maintained the same formulation, reaffirming centrality of shareholder interests statements such as: "The primary functions of the Honeywell International Inc. [...] Board Directors [...] are to oversee management of performance on behalf of the shareowners, to ensure that the long-term interests of the shareowners are being served" (p. 1). Likewise, The Home Depot, Inc. (2023), both before and after joining the BRT, has continued to emphasize that: "The Board of Directors of The Home Depot, Inc. is committed to maximizing long-term shareholder value while supporting management in the business and operations of the Company" (p. 1).

These examples demonstrate that for many companies, adherence to the BRT has not led to a substantial change in governance practices but rather a continuity with the shareholder primacy paradigm. The consistent use of expressions emphasizing shareholder value as a priority objective suggests that, in many cases, the alleged transition towards stakeholderism remains more of a formal than a substantive change.

Table 2. Excerpts demonstrating shareholder governance

Company	Excerpt
Abbott	"They must be first and foremost able and willing to represent the stockholders' short-term and long-term economic interests".
Apple Inc.	"The fundamental role of the directors is to exercise their business judgment to act in what they reasonably believe to be the best interests of the Corporation and its shareholders".
Chevron Corporation	"The Board Nominating and Governance Committee annually reviews the composition of the Board as a whole to assess the skills and characteristics that are currently represented on the Board, and in individual Directors, as well as the skills and characteristics that the Board may find valuable in the future, in light of the current and anticipated strategic plans and operating requirements of the Corporation and the long-term interests of stockholders".
Exxon Mobil Corporation	"The directors' fiduciary duty is to exercise their business judgment in the best interests of ExxonMobil's shareholders".
Honeywell	"The primary functions of the Honeywell International Inc. Board of Directors are to oversee management performance on behalf of the shareowners, to ensure that the long-term interests of the shareowners are being served".
Morgan Stanley	"Directors should also have the ability to represent the interests of shareholders".
The Home Depot, Inc.	"The Board of Directors of The Home Depot, Inc. is committed to maximizing long-term shareholder value while supporting management in the business and operations of the Company".
Procter & Gamble	"It is expected that Board members (in that role) will exercise diligently and in good faith their independent judgment in the best interests of the Company and its shareholders as a whole, notwithstanding their other activities or affiliations".

#### Source: Authors' elaboration.

# 4.2. Genuine transition or facade strategy?

contrary, some companies demonstrated a genuine shift in their governance approach. BlackRock, Inc., for instance, underwent a clear transition from 2020 to 2022, moving from explicitly shareholder-oriented language to approach that explicitly includes stakeholders among the beneficiaries of corporate decisions. In 2020, BlackRock, Inc.'s corporate governance guidelines stated: "The members of the Board are elected by the Company's shareholders and use their business judgment to direct, provide counsel and oversee the management of the Company in the interest of and for the benefit of the Company and its shareholders". However, in 2022 and in 2025, this statement was significantly modified to also include stakeholders: "The members of the Board are elected by the Company's shareholders and use their business judgment to direct, provide counsel and oversee the management of the Company in the interest of and for the benefit of the Company and its shareholders and other stakeholders" (BlackRock, Inc., 2025, p. 1). This transformation is particularly significant, considering that it occurred after the company's adherence to the BRT. This suggests that, at least in some cases, membership in the organization has acted as a catalyst for a more inclusive governance model. However, BlackRock, Inc.'s case remains an exception rather than the norm. Many companies that have joined the BRT continue to operate under a shareholder primacy logic, merely introducing linguistic adjustments in their governance guidelines without implementing a true strategic transformation. This phenomenon can be interpreted as an attempt to build an image of corporate social responsibility without making substantive

changes to governance practices. In some cases, the modifications introduced appear to be more of a facade operation rather than a genuine strategic transformation.

**Table 3.** Excerpts demonstrating a transition in governance

Company	Excerpt 2020	Excerpt 2022	Excerpt 2025
BlackRock, Inc.	"The members of the Board are elected by the Company's shareholders and use their business judgment to direct, provide counsel and oversee the management of the Company in the interest of and for the benefit of the Company and its shareholders".	"The members of the Board are elected by the Company's shareholders and use their business judgment to direct, provide counsel and oversee the management of the Company in the interest of and for the benefit of the Company and its shareholders and other stakeholders".	in the interest of and for the benefit of the Company and

Source: Authors' elaboration.

# 4.3. Hybrid models: Sustained evolution in Boeing and Citigroup Inc.

Another interesting aspect that emerged from the analysis concerns companies that had already consistently maintained or strengthened a hybrid governance model throughout the study period, balancing attention between shareholders and stakeholders. Boeing and Citigroup Inc. are clear examples of this trend: their corporate guidelines have consistently acknowledged the need to act in the interests of shareholders while simultaneously emphasizing the importance of considering other stakeholders, such as employees, customers, and communities. For instance, Citigroup Inc. (2025) states: "[...] take into account and the legitimate interests and concerns of all of Citi's stockholders and our other stakeholders in reaching decisions [...]" (p. 3). This demonstrates the company has historically embraced, continues to develop, a more inclusive approach compared to the traditional shareholder model, indicating a sustained evolution rather than a solely post-BRT shift.

With the update of their governance guidelines, these companies have further strengthened their

stakeholder-oriented approach, as evidenced by new formulations explicitly referencing the importance relationships with various stakeholders. For example, Boeing (2025), in its updated governance guidelines, states: "The Board and the officers recognize that the long-term interests of the Company and its shareholders are advanced when they take into account the concerns of employees, customers, suppliers, and communities" (p. 1). Additionally, it highlights that: "The CEO and other officers are responsible for establishing effective communications with the Company's stakeholders, including shareholders, employees, customers, suppliers, communities, governments, creditors, and corporate partners" (Boeing, 2025, p. 4). These statements indicate a clear, ongoing evolution in recognizing the role of stakeholders. This evolution suggests that, in some cases, the transition toward stakeholderism is not merely a facade but rather a gradual and sustained shift in governance strategies. The presence of these companies, which had already adopted a more balanced governance model before joining the BRT, reinforces the idea that the transition is underway, albeit with varying paces and nuances across all the analyzed firms.

Table 4. Hybrid models

Company	Excerpt
Boeing	"The Board and the officers recognize that the long-term interests of the Company and its shareholders are advanced when they take into account the concerns of employees, customers, suppliers, and communities".
Citigroup, Inc.	"[] take into account and balance the legitimate interests and concerns of all of Citi's stockholders and our other stakeholders in reaching decisions []".

Source: Authors' elaboration.

# 4.4. Language as an indicator of change

A particularly significant element emerging from the study concerns the use of language in governance guidelines, highlighting that 87.2% (34 out of 39) of companies adopt a shareholderoriented language, while only 12.8% (5 out of 39) stakeholder-oriented language. many companies maintain a shareholder-oriented governance model, there is a growing adoption of more inclusive language that emphasizes the role of stakeholders. A significant example is Walmart Inc., which, while remaining committed to a shareholderdriven governance approach, has introduced references to the importance of relationships with customers, suppliers, and local communities. Its governance guidelines state: "The Board acknowledges that the Company's long-term success

depends upon its strong relationship with its customers, associates, suppliers, and the communities, including the global community, in which it operates" (Walmart Inc., 2025, p. 3).

The same applies to S&P Global Inc. (2025), which, although not having altered its governance structure, has included in its guidelines statements centered on ESG issues, such as: "Monitoring the Corporation's strategy and material activities, practices, and policies related to environmental, social, and governance matters" (p. 1). Additionally, S&P Global Inc. (2025) emphasizes the importance of diversity within its board of directors with statements like: "The Board believes that its membership should reflect a diversity of occupational and personal backgrounds and experience, including gender, race, ethnicity, and age, to obtain a wide range of viewpoints and perspectives" (p. 2).

On the other hand, companies such as Lockheed Martin Corporation, Stryker, RTX Corporation, and JPMorgan Chase & Co. continue to use explicitly shareholder-oriented language, highlighting that for the vast majority of firms (34 out of 39), linguistic transformation has not yet occurred. For example, Lockheed Martin Corporation (2023) states: "The Board is responsible for succession planning for the Chief Executive Officer, ensuring leadership continuity that maximizes value for stockholders" (p. 9). Similarly, Stryker (2024) underscores the centrality of shareholders with statements like: "All non-employee directors are expected to have a meaningful share ownership position in the Company to reinforce the alignment of the interests of the Board and shareholders" (para. 1.10). RTX Corporation (2019) follows a similar approach, stating: "To further encourage the alignment of management and shareowner interests, the Board will, from time to time, adopt stock ownership requirements for non-management directors and the Company's Executive Leadership Group" (p. 2). Finally, JPMorgan Chase & Co. (2019) clarifies its shareholder-oriented stance with statements such as: "All directors should act in the best interests of the company and its shareholders" (p. 1).

The case of these companies demonstrates that adherence to the BRT has not yet led to

a structural shift in how governance is conceived and communicated.

The analysis of corporate governance guidelines among BRT companies reveals a complex and, in many cases, contradictory landscape. On one hand, there are signs of a gradual transition toward a more stakeholder-oriented governance model, with some companies (BlackRock, Inc., Boeing, and Citigroup Inc.) significantly updating their guidelines in both governance principles and language. On the other hand, the majority of firms (36 out of 39) continue to adhere to a strongly shareholder-oriented governance approach, merely adopting more inclusive language without making substantial changes to their practices.

This phenomenon suggests that the transition toward stakeholderism, while present, is not yet fully established and often appears more rhetorical than substantive. However, the presence of companies that have undertaken a genuine transformation, such as BlackRock, Inc., Boeing, and Citigroup Inc., leaves open the possibility that this trend could accelerate in the coming years. It will be interesting to observe how BRT firms continue to evolve and whether external pressures from investors, customers, and regulations will further push corporate governance toward a more inclusive and sustainable model.

**Table 4.** Linguistic analysis of corporate governance guidelines

Company	Excerpt		
JPMorgan Chase & Co.	"Directors are expected to act in the best interests of the Firm's shareholders".		
RTX Corporation	"To further encourage the alignment of management and shareowner interests, the Board will, from time to time, adopt stock ownership requirements for non-management directors and the Company's Executive Leadership Group".		
S&P Global Inc.	"Monitoring the Corporation's strategy and material activities, practices, and policies related to environmental, social, and governance ('ESG') matters".		
Stryker	"All non-employee directors are expected to have a meaningful share ownership position in the Company to reinforce the alignment of the interests of the Board and shareholders".		
Walmart, Inc.	"The Board acknowledges that the Company's long-term success depends upon its strong relationship with its customers, associates, suppliers, and the communities, including the global community, in which it operates".		
Lockheed Martin Corporation	"The Board is responsible for succession planning for the Chief Executive Officer, ensuring leadership continuity that maximizes value for stockholders".		

#### Source: Authors' elaboration.

#### 5. DISCUSSION OF THE RESULTS

This study examines the shift from shareholder primacy to stakeholder governance, highlighting both its practical implications and its rhetorical The analysis of corporate governance guidelines reveals a more complex reality than public declarations suggest. Empirical data indicate that most companies remain primarily focused on shareholders, reinforcing the persistence an entrenched paradigm. The research unveils a scenario in which corporate statements have adapted to social and regulatory pressures without a corresponding structural transformation in decision-making dynamics and governance models. While some companies have adopted more inclusive language, this has not translated into operational and structural mechanisms that genuinely balance the interests of stakeholders and shareholders. The following discussion rigorously and deeply explores the implications of these findings, providing a significant theoretical and empirical contribution to the governance debate. The emerging evidence not only clarifies the nature of the ongoing transformation but also challenges the actual effectiveness of corporate public statements, raising crucial questions about the future balance between shareholder and stakeholder-oriented interests.

Our study reveals a complex and, in many cases, contradictory reality between companies public statements and their actual governance practices. The examination of corporate governance guidelines highlights a marked divergence between public declarations in favor of stakeholderism and actual corporate management practices. Despite the 2019 BRT Statement advocating a shift toward more inclusive governance, our analyses indicate that many companies have not translated these intentions into significant structural changes. Decision-making priorities remain strongly anchored in shareholder value maximization, with language that continues to emphasize fiduciary duty toward shareholders. This discrepancy strengthens the hypothesis that, for many companies, adopting a stakeholder-oriented rhetoric serves more as a strategy for social and reputational legitimacy rather than a substantial change in governance dynamics. Although some firms have integrated stakeholder references into their official statements, the lack of concrete tools, such as redefining performance metrics or creating stakeholder representation bodies, suggests that the traditional paradigm of shareholder primacy continues to dominate corporate governance. This trend underscores the importance of critical monitoring and stricter regulation to ensure that stakeholderism does not remain confined to a purely rhetorical level.

The persistence of shareholder-oriented language in governance documents is particularly telling. While some companies have incorporated stakeholder references, these additions appear more rhetorical than operational. The absence of concrete mechanisms, such as stakeholder representation on boards, stakeholder-linked performance metrics, or revisions to executive compensation structures, indicates that stakeholder governance largely remains an aspiration rather than a codified governance principle.

Our findings strongly support hypothesis H1, which states that companies continue to follow a shareholder-oriented governance model. Almost all of the companies analysed (38 out of 39) maintain governance frameworks that prioritise the maximisation of shareholder value. This is evident in the explicit statements found in corporate governance guidelines, where fiduciary duty to shareholders remains central. Companies such as Apple Inc., Morgan Stanley, Abbott Laboratories, and The Home Depot, Inc. illustrate this trend, as their governance policies continue to shareholder interests without substantial references to stakeholders. For example, Apple Inc.'s statement: "The Board oversees the Chief Executive Officer [...] and seeks to ensure that the long-term interests of shareholders are being served" (p. 1). Similarly, Honeywell has consistently reaffirmed the centrality of shareholder interests.

This result aligns with the work of Bebchuk and Tallarita (2020), who argue that companies have not implemented structural changes following the BRT statement. Instead, their adherence to shareholder primacy remains intact, with governance structures and executive incentives still heavily tied to financial performance metrics. The persistence of this paradigm suggests that, despite growing external pressures, companies are reluctant to deviate from established governance norms that prioritize shareholder returns. This persistence suggests that, for almost all companies (38 out of 39), adherence to the BRT has not led to a substantial change in governance practices, but rather a continuation of the shareholder primacy paradigm. The consistent use of expressions emphasising shareholder value as a priority objective indicates that the supposed transition to stakeholderism remains more of a formal change than a substantial transformation.

Although there is some evidence of a shift toward stakeholderism, our findings indicate that this transition is far from universal; consequently, we do not support hypothesis H2, which states that companies are progressively adopting a stakeholder-oriented governance model across the board. Companies like BlackRock, Inc., and Citigroup Inc. have made significant changes to their governance language and structures, incorporating explicit references to stakeholder considerations. BlackRock, Inc., for instance, has revised its governance guidelines to acknowledge the importance of stakeholders beyond shareholders, signalling a possible evolution in governance philosophy. BlackRock, Inc.'s guidelines have shifted explicitly shareholder-oriented language in 2020 to explicitly including stakeholders among the beneficiaries of corporate decisions in 2022 and 2024. This transformation is particularly relevant as it occurred after the company joined the BRT, suggesting that membership in the organization acted as a catalyst for a more inclusive governance model, albeit an exception rather than the norm.

However, these cases remain exceptions rather than the norm. The broader corporate landscape continues to reflect a strong shareholder orientation, with many firms using stakeholder rhetoric without implementing meaningful governance reforms. For most companies (36 out of 39) that joined the BRT, there has been a continuation of the logic of shareholder primacy, with the introduction of mere linguistic changes in governance guidelines, without any real strategic transformation. This phenomenon can be interpreted as an attempt to build an image of corporate social responsibility, or "governance washing" (Esposito De Falco, 2022), without substantially altering governance practices. The changes introduced appear more like window dressing than a genuine strategic transition. This suggests that, while stakeholderism is gaining traction in public discourse, its practical adoption remains limited. The increasing regulatory focus on ESG factors and investor pressures may accelerate this shift in the future, but for now, the transition appears more incremental than transformative.

Our findings provide partial support for hypothesis H3, which posits a movement towards a hybrid governance model in which economic and financial value creation and stakeholder considerations coexist in a strategic and sustainable long-term balance. Some companies, such as Boeing and Citigroup Inc., demonstrate a hybrid approach, stakeholder considerations integrating maintaining a strong shareholder orientation. These companies had already adopted a hybrid governance model before joining the BRT. Their corporate guidelines have always recognised the need to act in the interests of shareholders, while emphasising the importance of considering other stakeholders such as employees, customers, and communities. With the update of their governance guidelines, they have further strengthened their stakeholder-oriented approach, as demonstrated by new wording that explicitly refers to the importance of relationships with various stakeholders.

These companies recognize the importance of stakeholder relationships for long-term value creation but have not fully institutionalized stakeholder governance. This hybrid model aligns with emerging research suggesting that companies may adopt a dual-focus approach, acknowledging the need to balance financial performance with broader social and environmental concerns (Freeman et al., 2020). The integration of ESG considerations into governance frameworks further reinforces this perspective, as companies seek to mitigate reputational risks and respond to evolving investor expectations. This suggests that, in some cases, the transition to stakeholderism is not merely a facade, but rather a gradual shift in governance strategies. However, these hybrid models are not widespread among most companies, and for most firms, these efforts remain supplementary rather than fundamental to their governance strategies.

A significant finding of the study concerns the use of language in governance guidelines. Our analysis shows that 34 out of 39 companies adopt shareholder-oriented language, while only five use stakeholder-oriented language. Although many companies maintain a shareholder-focused governance model, there is a growing adoption of more inclusive language that emphasises the role of stakeholders. For example, Walmart Inc., while maintaining a shareholder-oriented governance approach, has introduced references to the importance of

relationships with customers, suppliers, and local communities. Similarly, S&P Global Inc. has included statements focused on ESG issues in its guidelines.

On the other hand, companies such as Lockheed Martin Corporation, Stryker, Corporation, and JPMorgan Chase & Co. continue to explicitly shareholder-oriented language. For example, Lockheed Martin Corporation (2025) states: "The Board is responsible for succession planning for the Chief Executive Officer" (p. 9), ensuring leadership continuity that maximises value for stockholders. This highlights that for companies (34 out of 39), the linguistic transformation has not yet taken place, and adherence to the BRT has not led to a structural change in the way governance is conceived and communicated. The mere adoption of more inclusive language without substantial changes in practices suggests that the transition to stakeholderism, although present, is not yet fully consolidated and often appears more rhetorical than substantial.

## 6. CONCLUSION

This study examined the transition from shareholder primacy to stakeholder governance among companies that signed the 2019 BRT Statement, providing an updated empirical analysis of their corporate governance practices.

Our research answers the research question and hypotheses, demonstrating that adherence to the BRT remains, for most companies, largely symbolic. The findings highlight a persistent discrepancy between public rhetoric in favour of stakeholderism and actual corporate management practices.

The analysis reveals strong support for hypothesis H1, as 38 out of 39 companies continue to maintain governance models that prioritise shareholder value maximisation. This is corroborated by an examination of their guidelines, which reaffirm the centrality of the fiduciary duty to shareholders.

Hypothesis H2, which predicted the progressive and widespread adoption of a stakeholder-oriented governance model, was not universally supported by our findings. Although a few exceptions, including BlackRock, Inc., Boeing, and Citigroup Inc., have been identified that have undertaken genuine transformation or adopted hybrid models, most companies (36 out of 39) continue to operate with a strong shareholder orientation, despite adopting more inclusive language.

Our findings offer partial support for hypothesis *H3*, which suggests a move towards a hybrid governance model. Companies such as Boeing and Citigroup Inc. demonstrate an approach that balances stakeholder considerations with a strong shareholder orientation. However, for most companies, these efforts remain complementary and not fundamental to their governance strategies.

In summary, our findings confirm that the governance guidelines of BRT signatory companies have not undergone structural transformations in favour of greater stakeholder involvement in corporate strategies. This suggests that stakeholderism is often used more as a marketing tool than as a genuine evolution of the governance model.

This study provides a crucial contribution to the corporate governance debate by highlighting the critical challenges and practical implications of the transition from a shareholder primacydominated model to a more inclusive one. From a theoretical perspective, the research underscores how stakeholderism is often adopted more as communicative tool than as a genuine transformation of the decision-making paradigm. Our analysis demonstrates that many companies maintain structural and decision-making frameworks that predominantly favour shareholders, leaving stakeholders in a secondary position. persistence indicates that change cannot be limited to a mere revision of corporate narratives but requires a radical rethinking of governance mechanisms and the adoption of concrete tools to ensure the effective inclusion of stakeholders in decision-making processes. Only through structural modifications, such as greater transparency in defining corporate objectives and better alignment between managerial incentives and social impacts, will it be possible to overcome the dichotomy between principled statements and actual practices. The shift toward a truly stakeholder-oriented governance model cannot remain a symbolic evolution but must translate into a tangible transformation of corporate management.

From a practical standpoint, the findings of this research have significant implications for various stakeholders. For policymakers, there is an urgent need for more effective regulatory instruments to transform principled declarations into concrete actions, promoting governance that genuinely balances financial and social interests. This includes the need for reforms that ensure greater accountability and transparency in corporate strategies, going beyond rhetoric to impose mechanisms such as stakeholder representation on boards, stakeholder-related performance metrics, and reviews of executive compensation structures, which are currently lacking in most companies. The regulatory framework could draw inspiration from more rigorous principles, such as those of the OECD, which Samanta (2019) used to measure convergence towards shareholder primacy in national regulations, thus suggesting areas where new regulations could push towards stakeholderism.

For institutional investors, our study provides a valuable framework for assessing the authenticity of corporate commitments to sustainability and social responsibility, emphasizing the need for more rigorous metrics to monitor actual governance practices (Earley, 2016). We emphasise the need for more rigorous metrics to monitor actual governance practices, distinguishing between mere statements and real operational transformations. For corporate leaders, this research highlights the importance of adopting strategies that go beyond mere regulatory compliance, credibly and measurably integrating stakeholder interests into decision-making processes. This implies a commitment to transparency and the implementation of mechanisms that reflect a genuine balance between shareholder value and stakeholder impact.

Our study presents several limitations that pave the way for future research. The analysis was based on an examination of corporate governance guidelines, which, while an important source of information, do not necessarily reflect the full spectrum of corporate practices. This qualitative approach, while useful for capturing nuances in language, did not directly explore actual corporate decisions or the economic and social impact of governance strategies. A more in-depth investigation

should include interviews with corporate executives, analysis of actual business behaviors, and comparative evaluations across different jurisdictions provide a more comprehensive understanding of stakeholderism's real implementation.

The nature of the metrics used to assess stakeholder governance orientation represents another limitation. Measuring corporate transformation requires more sophisticated tools that go beyond textual analysis and integrate empirical data on corporate decisions, value distribution among different stakeholders, and the social and economic impact of governance strategies. Future studies should develop more sophisticated tools that integrate empirical data on business decisions, value distribution among different stakeholders, and the social and economic impact of governance strategies. Employing quantitative methods akin to the leximetric analysis by Samanta (2019) or developing more detailed evaluation frameworks based on operationalized ESG metrics could offer more granular insights into corporate transformation.

Future quantitative extensions of this research could leverage established parameters, similar to Samanta's (2019) methodology, to further validate quantify the observed shifts towards shareholder primacy or stakeholder integration by categorizing governance features based on their compulsory nature, optionality, or absence. This could provide a comprehensive lens through which to analyze how legal and quasi-legal norms evolve to reflect prevailing governance philosophies.

Future research should include a more detailed investigation incorporating interviews with company executives, analysis of actual corporate behavior, and comparative assessments across jurisdictions to gain a more comprehensive understanding of the actual implementation of stakeholderism. It will be crucial to develop more detailed assessment frameworks to measure the degree of stakeholderism adoption in large corporations. Furthermore, exploring the extent to which regulatory divergences between countries, such as those highlighted by Samanta (2019) in the context of convergence shareholder primacy in regulations, towards influence the adoption or non-adoption of stakeholder governance policies at the corporate level would provide valuable insights.

Exploring the qualitative reasons behind observed divergence or stagnation stakeholder-oriented practices post-2008 financial crisis, as indicated by broader corporate governance trends, would be valuable. This includes analysing the impact of specific regulatory changes, investor activism, and societal pressures on actual corporate behavior, rather than solely formal disclosures.

Our study highlights the need for a more critical and systematic analysis of corporate statements of principle and their actual governance practices. Through continuous monitoring and the adoption of more rigorous measurement tools, it will be possible to distinguish between genuine commitment and mere communication strategies. Whether the future of corporate governance will be truly stakeholder-oriented will depend on the ability of companies to translate rhetoric into action and on institutions to create a regulatory framework that rewards the most inclusive and sustainable practices. The evolution of the stakeholderism debate in the coming years will therefore be crucial to understanding whether this paradigm can become a new standard or remain an unfulfilled promise.

#### REFERENCES

- Amis, J., Barney, J., Mahoney, J. T., & Wang, H. (2020). From the editors Why we need a theory of stakeholder governance and why this is a hard problem. Academy of Management Review, 45(3), 499-503. https://doi.org/10.5465/amr.2020.0181
- Apple Inc. (2025, August 19). Corporate governance guidelines. https://s2.q4cdn.com/470004039/files/doc \_downloads/Governance/Corporate-Governance-Guidelines.pdf
- Attanasio, G., Preghenella, N., De Toni, A. F., & Battistella, C. (2022). Stakeholder engagement in business models for sustainability: The stakeholder value flow model for sustainable development. *Business Strategy and* the Environment, 31(3), 860-874. https://doi.org/10.1002/bse.2922
- Baah, C., Jin, Z., & Tang, L. (2020). Organizational and regulatory stakeholder pressures friends or foes to green logistics practices and financial performance: Investigating corporate reputation as a missing link. Journal
- of Cleaner Production, 247, 119-125. https://doi.org/10.1016/j.jclepro.2019.119125
  Bebchuk, L. A., & Tallarita, R. (2020). The illusory promise of stakeholder governance. Cornell Law Review, 106, 91-177. https://publications.lawschool.cornell.edu/lawreview/wp-content/uploads/sites/2/2021/02/The-Illusory -Promise-of-Stakeholder-Governance.pdf
- Becht, M., Jenkinson, T., & Mayer, C. (2005). Corporate governance: An assessment. Oxford Review of Economic Policy, 21(2), 155-163. https://doi.org/10.1093/oxrep/gri009
- Bhandari, K. R., Ranta, M., & Salo, J. (2022). The resource-based view, stakeholder capitalism, ESG, and sustainable competitive advantage: The firm's embeddedness into ecology, society, and governance. Business Strategy and the Environment, 31(4), 1525-1537. https://doi.org/10.1002/bse.2967
- BlackRock, Inc. (2025, March 27). Corporate governance guidelines. https://s24.q4cdn.com/856567660/files/doc \_governance/2025/Apr/09/Corporate-Governance-Guidelines-2025-\_vFF-b44dee.pdf
- Boeing. (2025, April 24). Corporate governance principles. https://www.boeing.com/content/dam/boeing/boeingdotcom /company/general\_info/pdf/corporate-governance-principles.pdf
- Braun, C., & Busuioc, M. (2020). Stakeholder engagement as a conduit for regulatory legitimacy? Journal of European Public Policy, 27(11), 1599-1611. https://doi.org/10.1080/13501763.2020.1817133
- Citigroup Inc. (2025, August 1). Corporate governance guidelines. https://www.citigroup.com/rcs/citigpa/storage
- /public/citigroup-corporate-governance-guidelines.pdf
  Collins, H. (2001). Regulating the employment relation for competitiveness. *Industrial Law Journal*, 30(1), 17-48. https://doi.org/10.1093/ilj/30.1.17
- Dias, G. N., Hamza, K. M., Lievens, A., & Moons, I. (2025). Sustainability-based value creation within a multistakeholder network: Balancing expectation conflicts within the Amazon context. Business Strategy and the Environment, 34(1), 116-128. https://doi.org/10.1002/bse.3981

- Dmytriyev, S. D., Freeman, R. E., & Hörisch, J. (2021). The relationship between stakeholder theory and corporate social responsibility: Differences, similarities, and implications for social issues in management. Journal of Management Studies, 58(6), 1441-1470. https://doi.org/10.1111/joms.12684
- Donaldson, T., & Preston, L. E. (1995). The stakeholder theory of the corporation: Concepts, evidence, and implications. *Academy of Management Review*, 20(1), 65-91. https://doi.org/10.5465/amr.1995 .9503271992
- Earley, S. (2016). Metrics-driven information governance. IT Professional, 18(2), 17-21. https://doi.org/10.1109 /MITP.2016.26
- Easterbrook, F. H., & Fischel, D. R. (1996). *The economic structure of corporate law*. Harvard University Press. Eccles, R. G., & Klimenko, S. (2019). The investor revolution. *Harvard Business Review*, *97*(3), 106–116. https://hbr.org/2019/05/the-investor-revolution
- Edmans, A. (2011). Does the stock market fully value intangibles? Employee satisfaction and equity prices. Journal of Financial Economics, 101(3), 621-640. https://doi.org/10.1016/j.jfineco.2011.03.021
- Endo, K. (2020). Corporate governance beyond the shareholder-stakeholder dichotomy: Lessons from Japanese corporations' environmental performance. Business Strategy and the Environment, 29(4), 1625-1633. https://doi.org/10.1002/bse.2457
- Eng, L. L., Fikru, M., & Vichitsarawong, T. (2022). Comparing the informativeness of sustainability disclosures versus ESG disclosure ratings. *Sustainability Accounting, Management and Policy Journal*, 13(2), 494–518. https://doi.org/10.1108/SAMPJ-03-2021-0095
- Esposito De Falco, S. (2022). Dalla corporate governance alla sustainable governance washing [From corporate governance to sustainable governance washing]. *Corporate Governance and Research & Development Studies*, 2, 11-13. https://doi.org/10.3280/cgrds2-20210a13207
- Esposito De Falco, S. (2024). Corporate governance: Teorie, attori e sistemi di controllo nelle imprese [Corporate governance: Theories, actors, and control systems in companies]. McGraw-Hill.
- Fransen, L. (2012). Multi-stakeholder governance and voluntary programme interactions: Legitimation politics in the institutional design of corporate social responsibility. *Socio-Economic Review*, 10(1), 163–192. https://doi.org/10.1093/ser/mwr029
- Freeman, R. E. (2010). Strategic management: A stakeholder approach. Cambridge University Press. https://doi.org /10.1017/CBO9781139192675
- Freeman, R. E., Dmytriyev, S. D., & Phillips, R. A. (2021). Stakeholder theory and the resource-based view of the firm. *Journal of Management*, 47(7), 1757–1770. https://doi.org/10.1177/0149206321993576

  Freeman, R. E., Phillips, R., & Sisodia, R. (2020). Tensions in stakeholder theory. *Business & Society*, 59(2), 213–231.
- https://doi.org/10.1177/0007650318773750
- Gupta, S., & Aggarwal, D. (2024). Shrinking the capital costs and beta risk impediments through ESG: Study of an emerging market. Asian Review of Accounting, 32(2), 249–277. https://doi.org/10.1108/ARA-05-2023-0130
- Hall, J., & Vredenburg, H. (2005, October 1). Managing stakeholder ambiguity. MIT Sloan Management Review, 47(1). https://sloanreview.mit.edu/article/predicting-customer-choices-2/
- Paper No. 521/2017). European Corporate Governance Institute (ECGI). https://doi.org/10.2139/ssrn.3004794 Hart, O., & Zingales, L. (2017). Companies should maximize shareholder welfare not market value (Finance Working
- Heeb, F., Kölbel, J. F., Paetzold, F., & Zeisberger, S. (2023). Do investors care about impact? The Review of Financial Studies, 36(5), 1737-1787. https://doi.org/10.1093/rfs/hhac066
- Honeywell International Inc. (2025, March 25). Corporate governance guidelines. https://investor.honeywell.com static-files/0deee97c-3801-47cb-ae74-e4648578802b
- Jalilvand, M. A., Raeisi, A. R., & Shaarbafchizadeh, N. (2024). Accountable hospital governance structure rules in Iran: A qualitative document analysis. International Journal of Health Governance, 29(3), 271-283. https://doi.org/10.1108/IJHG-03-2024-0022
- Jang, G.-Y., Kang, H.-G., & Kim, W. (2022). Corporate executives' incentives and ESG performance. Finance Research Letters, 49, Article 103187. https://doi.org/10.1016/j.frl.2022.103187
- Jensen, M. C. (2002). Value maximization, stakeholder theory, and the corporate objective function. *Business Ethics Quarterly*, 12(2), 235–256. https://doi.org/10.2307/3857812
- Jensen, M. C., & Meckling, W. H. (1976). Theory of the firm: Managerial behavior, agency costs and ownership structure. Journal of Financial Economics, 3(4), 305-360. https://doi.org/10.1016/0304-405X(76)90026-X
- Joshi, A. M., Antons, D., Piening, E. P., Dienhart, C., & Salge, T. (2024). Governing transnational commons: How international treaties and multi-stakeholder organizations shape cooperation and conflict. *Journal of Management Studies*. Advance online publication. https://doi.org/10.1111/joms.13165
- JPMorgan Chase & Co. (2019, February). Corporate governance principles and proxy voting guidelines. https://am.jpmorgan.com/content/dam/jpm-am-aem/asiapacific/hk/en/corporate-governance-pdf/CGVPG 1603.pdf
- Kramer, M. R., & Porter, M. R. (2011). Creating shared value. *Harvard Business Review*, *17*(1–2), 62–77. https://hbr.org/2011/01/the-big-idea-creating-shared-value
- Kyaw, K., Olugbode, M., & Petracci, B. (2022). Stakeholder engagement: Investors' environmental risk aversion and corporate earnings. Business Strategy and the Environment, 31(3), 1220-1231. https://doi.org/10.1002 /bse.2951
- Lazonick, W. (2014). Profits without prosperity: Stock buybacks manipulate the market and leave most Americans worse off. Harvard Business Review, 92(9), 46-55. https://hbr.org/2014/09/profits-without-prosperity
- Lockheed Martin Corporation. (2023, June 22). Corporate governance quidelines. https://www.lockheedmartin.com content/dam/lockheed-martin/eo/documents/leadership/corporate-governance-guidelines.pdf
- Lockheed Martin Corporation. (2025). Proxy statement & notice of annual meeting. https://www.lockheedmartin .com/content/dam/lockheed-martin/eo/documents/annual-reports/2025-proxy-statement.pdf
- Luttmer, E. G. J. (2007). Selection, growth, and the size distribution of firms. The Quarterly Journal of Economics, 122(3), 1103-1144. https://doi.org/10.1162/qjec.122.3.1103
- Martin, K., & Phillips, R. (2022). Stakeholder friction. Journal of Business Ethics, 177(3), 519-531. https://doi.org/10 .1007/s10551-020-04652-9
- Morgan Stanley. (2025, March 31). Morgan Stanley board of directors corporate governance policies. https://www.morganstanley.com/about-us-governance/cgpolicies

- Organization for Economic Co-operation and Development (OECD). (2004). OECD principles of corporate governance. https://www.oecd.org/content/dam/oecd/en/publications/reports/2004/05/oecd-principles-of-corporategovernance-2004\_g1gh44a5/9789264015999-en.pdf
- Ortas, E., Gallego-Álvarez, I., & Alvarez, I. (2019). National institutions, stakeholder engagement, and firms' environmental, social, and governance performance. Corporate Social Responsibility and Environmental Management, 26(3), 598-611. https://doi.org/10.1002/csr.1706
- Ostapenko, V., Tyshchenko, V., Rats, O., & Brusentseva, O. (2024). Structuring and determining the influence of stakeholders on the innovation ecosystem. *Journal of Governance & Regulation*, 13(1), 123–134. https://doi.org/10.22495/jgrv13i1art11
- Quinto, I., Scandurra, G., & Thomas, A. (2019). Internal resources and stakeholders engagement affecting environmental innovations: An exploratory research. Corporate Governance and Research & Development Studies, 1, 59-86. https://doi.org/10.3280/cgrds1-2019oa8609
- Rautenstrauch, T., & Hummel, J. (2022). Shareholder activism: A Swiss case study. Corporate Ownership & Control, 19(4), 30-41. https://doi.org/10.22495/cocv19i4art3
- Raytheon Technologies Corporation (RTX Corporation). (2019, April). Corporate governance guidelines. https://prdsc102-cdn.rtx.com/-/media/rtx/r/rtx-corporate-governance/rtx-corporate-governance-guidelines.pdf?rev =b26380aefd0f482ab6a3786e66d9b5d9
- S&P Global Inc. (2025, May 6). Corporate governance guidelines. https://s29.q4cdn.com/690959130/files/doc\_governance/2025/S-P-Global-Corporate-Governance-Guidelines-May-2025.pdf
- Samanta, N. (2019). Convergence to shareholder primacy corporate governance: Evidence from a leximetric analysis of the evolution of corporate governance regulations in 21 countries, 1995-2014. Corporate Governance, 19(5), 849-883. https://doi.org/10.1108/CG-07-2018-0249
- Schaltegger, S., Hörisch, J., & Freeman, R. E. (2019). Business cases for sustainability: A stakeholder theory perspective. *Organization & Environment*, *32*(3), 191–212. https://doi.org/10.1177/1086026617722882
- Shah, S. N. U. H., & Li, Y. (2025). Shareholders engagement and annual general meetings of Australian listed companies: An empirical analysis of corporate governance. Journal of Governance & Regulation, 14(1), 206-217. https://doi.org/10.22495/jgrv14i1art19
- Shahzad, A. M., Rutherford, M. A., & Sharfman, M. P. (2016). Stakeholder-centric governance and corporate social performance: A cross-national study. *Corporate Social Responsibility and Environmental Management, 23*(2), 100-112. https://doi.org/10.1002/csr.1368
- Shleifer, A., & Vishny, R. W. (1997). A survey of corporate governance. The Journal of Finance, 52(2), 737-783. https://doi.org/10.1111/j.1540-6261.1997.tb04820.x
- Sternberg, E. (1997). The defects of stakeholder theory. Corporate Governance: An International Review, 5(1), 3-10. https://doi.org/10.1111/1467-8683.00034
- shareholder value Stout, L. A. (2024,October 5). The myth. The European Financial Review. https://www.europeanfinancialreview.com/the-shareholder-value-myth-2/
- (2024, May 8). Corporate governance guidelines. https://www.stryker.com/it/en/about/governance /corporate-governance-guidelines.html
- The Home Depot, Inc. (2023, November 16). Corporate governance guidelines of The Home Depot, Inc. board of directors. https://shorturl.at/hG8de
- Thrall, C. (2021). Public-private governance initiatives and corporate responses to stakeholder complaints.
- International Organization, 75(3), 803–836. https://doi.org/10.1017/S0020818321000199
  Tibiletti, V., Marchini, P. L., Furlotti, K., & Medioli, A. (2021). Does corporate governance matter in corporate social responsibility disclosure? Evidence from Italy in the "era of sustainability". Corporate Social Responsibility and Environmental Management, 28(2), 896-907. https://doi.org/10.1002/csr.2097
- Tirole, J. (2010). The theory of corporate finance. Princeton University Press.
- Vural-Yavaş, Ç. (2021). Economic policy uncertainty, stakeholder engagement, and environmental, social, and governance practices: The moderating effect of competition. *Corporate Social Responsibility and Environmental Management, 28*(1), 82–102. https://doi.org/10.1002/csr.2034

  Walmart Inc. (2025, June 5). *Corporate governance guidelines*. https://stock.walmart.com/\_assets/\_5116d3f16a 30802a54bfb9ec93948432/walmart/db/928/9547/file/Corporate+Governance+Guidelines.pdf
- Yoshikawa, T., Nippa, M., & Chua, G. (2021). Global shift towards stakeholder-oriented corporate governance: Evidence from the scholarly literature and future research opportunities. Multinational Business Review, 29(3), 321-347. https://doi.org/10.1108/MBR-10-2020-0200
- Zaman, R., Jain, T., Samara, G., & Jamali, D. (2022). Corporate governance meets corporate social responsibility: Mapping the interface. *Business & Society*, *61*(3), 690–752. https://doi.org/10.1177/0007650320973415